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Highlights:

We completed the baking of the gun chamber to drive out hydrogen.  The chamber is now cooling down.  We also ran the gun high voltage power supply up to > 100 kV in open air.  It is ready for assembly with the rest of the system.  Work continues on the Laser Personnel Safety System. We held a status review of the AES cryomodule assembly.

Management:
We continued to prepare the cost book and presentations for the FSU design review at the end of next week.
Gwyn Williams is off to England for a 4GLS User meeting and science review.

Discussions are underway considering possible JTO proposals.

We continue to prepare Work Package inputs for next FY's effort.

Tech Transfer:

We participated in a teleconference with Phillips Research exploring commercial possibilities of THz sources, such as medical and pharmaceutical applications.
Plans were finalized for next week’s visit by a delegation from the VEDP (Virginia Economic Development Partnership).

Also completed the Tech Transfer Annual Work Plan (AWP), incorporating Intellectual Property and Technology Transfer into one AWP.
 

RF:

RF – The setup has started for testing cryomodule windows using a 110 kW Injector klystron.  All klystrons will be put into HV next week to maintain their vacuum during the down.
 

Gun HVPS – The HVPS for the GTS Gun was operated to 150 kV with no load, to 5.1 mA into a short circuit, and was calibrated to readout directly in ma on the front panel meter.  SF6 is required for additional testing at higher voltages.

Instrumentation and Controls:

   Much effort has been dedicated to the User Labs' LPSS.  We are shaking out User Lab 1 currently and also progressing nicely with Lab 2.  Both of these Lab's LPSS boxes have been gone through and configured with the updated requirements including the speaker system and the expansions for hutches within the labs.  Two Hutch Interface status boxes were mounted in Labs 1 and 2.  Several other Hutch Interface status enclosures were fabricated for the other user labs when the modifications begin in those.  The documentation for these enclosures is being generated and maintained.  Several cables for communication between Lab 2 and the master LPSS PLC were installed along with some cabling terminations in Lab 6.  The documentation for all of these labs is being updated as changes and installations are made also hardware within the labs are being labeled with a standard layout for consistency and easy identification.  We are also in the process of replacing eight green status panel LED's in control room so remote status is readily available and accurate. 
    Progress continued this week on the High Voltage Power Supply Controller for the GTS.  The Analog card layout has been completed and is currently out for error checking and review.  This card will be used to control the HVPS Analog I/O as well as monitor the current of the electron beam dumps.  We've also agreed upon the interface for this chassis to the PSS system and the power supply so that all parties are satisfied.  We will have independent control over the power supply based upon our interlock chain as well as the PSS A & B status.  The PSS group will have discrete control over the power supply as well as the AC power contactor.  This provides both groups with their requirements and interlocks the power supply by three different methods.

    We continue to work on the VME interface for the timer cards and the BLM interface.  We've gotten the Varian Ion Pumps communicating with the IOC via RS-485 and the first draft EPICS interface is completed.  We will use these in the GTS as well to provide diagnostic and interlocks for the vacuum system.  The Lasermetrics Pulse Generators needed for the GTS Drive Laser are being modified to use safe high voltage connectors.  This will also allow us to use our standard cables that we already have available.  The enclosures for the Cathode Drive Motor Interface have been completed.  The relays to perform the limit switch logic have been installed and these boxes are ready for testing when time permits.
    More effort went this week updating the wiki further.  Component pages for the beam viewers and the quad magnets have been created.  These component pages have also been added to table pages dedicated for the beam viewers, quad magnets, vacuum ion pumps, beam position monitors, beam line valves, beam loss monitors, corrector magnets.  A help page to use as a reference in how to edit the tables so that they can be used like the Devlore page as far as changing status, comments and the flags of specific items on the FEL was created.     

   Individual pages for each component in the FEL are currently being generated so pictures can be linked to each item.  This will provide us with some visual information and give a brief description of each item’s functionality.

Beam Transport:

Improvements and Upgrades
Gun Test Stand (GTS)

• 
The design package per ASME Code Section VIII design standards for the SF6 Tank system is in peer review by Robby Hicks. He will do the finite element calculation on the intersection with the gun vacuum tank using ANSYS Workbench.  The drawings are in back-check to become “shop” drawings for our shop’s activities. The fabrication documentation package that will satisfy the requirements of the JLab pressure vessel committee for maintaining records of material certification, the welding per a qualified procedure and the maintenance of a material tracking system are in preparation. Meanwhile, the shell material came in and is rolled into half shells.  The weld samples to qualify the procedure are welded (by William Berkeley – the welder who will perform the welds on the tanks).  Advex Co. is testing the samples per code procedures.

• 
The Gun Solenoid remains in fabrication at New England Technicoil.

• 
Matthew Marchlik and Brian Hankla continued work on the SF6 Gas Transfer System getting most of the materials in.  We will do a final review on Monday before actual installation begins.

• 
The simplified Brewster window housing is in fabrication.  

•  
The design of Drive Laser Table system with a hutch continues.

•
The 400 C bake of the gun tank for the Gun Test Stand was completed successfully using the makeshift oven in the North access building.  The simplified oven design with shielded Cal rod heaters and no external blower was successful in keeping the tank uniform within 10 C during the bake. 

SF Sextupole Magnet Testing 
• 
A student in Magnet Test has worked out the power supply interface to the magnet measurement apparatus while on Student funds.  This advance will lead to the next part of the program, running a Sextupole up to field (again using student funds) while using variable ramp rates. This exercise will determine the rate limit (if any) at which residual fields from eddy currents may throw the Sextupole fields out of specification.  

Magnetic anomalies 
• 
No work this week
National High Field Magnet Laboratory FEL
• 
I was able to generate a high level cost estimate for this project using the data from the IR Demo with inflation escalations.

Terahertz: 
This week, the final plans for numerous upcoming studies at Brookhaven have been made.  Beam time on the U12 and U4IR beamlines has been confirmed, and Larry Carr has begun tests of Jiufeng Tu’s 10 T superconducting magnet now in place at U4IR.  The tests include the vacuum systems for transporting the beam through the magnet, which were designed and built here at JLab.  The specs of the magnet and optical system were discussed w/our collaborators, John Singleton and Susan Cox of the NHMFL at Los Alamos.  We will be performing THz spectroscopy on several materials to look for cyclotron resonance signatures.  John and Susan will spend some time determining the best materials to examine during our beam time on U4IR.  These studies are in support of the FSU-FEL project.  On U12, we will be performing several other reflectance and transmittance THz spectroscopy studies on several different materials.

Also this week, we contacted Thermo Nicolet regarding ordering replacement parts for the vacuum spectrometer system.  One of our globar sources has partially burned out so we currently have no spare.  The globar source is necessary for any mid IR studies we hope to do, such as Harvey Rutt’s double resonance experiment, as well as for performing periodic calibration tests on the spectrometer.
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